Author's response to reviews

Title: Dynamics of Enterococcus faecium Clones Colonizing Hospitalized Patients: data from a prospective observational study

Authors:

Maja Weisser (weisserm@uhbs.ch)
Evelien A Oostdijk (E.A.N.Oostdijk@umcutrecht.nl)
Rob JL Willems (rwillems@umcutrecht.nl)
Marc JM Bonten (m.j.m.bonten@umcutrecht.nl)
Reno Frei (freir@uhbs.ch)
Luigia Elzi (elzil@uhbs.ch)
Jörg F Halter (halterj@uhbs.ch)
Andreas F Widmer (widmera@uhbs.ch)
Janetta Top (J.Top@umcutrecht.nl)

Version: 4 Date: 22 February 2012

Author's response to reviews: see over
Resubmission Manuscript: “Dynamics of Ampicillin-resistant Enterococcus faecium Clones Colonizing Hospitalized Patients: Data from a Prospective Observational Study”.

Dear Editor, dear Dr. Gomes

We thank you and the reviewers for acceptance of our article. We are very happy that it will be published in the BMC Infectious Diseases. Please find enclosed the manuscript including the name of the Ethical Committee as requested. Formatting was controlled and is hopefully fulfilling all requirements.

Sincerely yours,
Dr. Maja Weisser
mweisser@uhbs.ch